Author's response to reviews

Title: Sublingual epidermoid cyst: a case report

Authors:

Tolga Kandogan (tkandogan@gmail.com)
Murat Koc (muratkoc@yahoo.com)
Enver Vardar (evardar@yahoo.com)
Elif Selek (elif@yahoo.com)
Ozlem Sezgin (osezgin@gmail.com)

Version: 4 Date: 11 August 2007

Author's response to reviews: see over
Comments to the reviewers:

Reviewer: Giorgio Iannetti

1

Q=Sublingual Epidermoid Cyst is an unusual localization of the disease, but in your case report is not clear why you performed thyroid scintigraphy as first step of the diagnostic process.

A= thyroid scintigraphy is not performed as first step of the diagnostic process. It was performed after MRI, and explained why it was performed, although it is not included to the algorithm.

2

Surgical technique was explained more extensively.

Reviewer: Adriano Piattelli

1

One of the mentioned references is included to the reference list.


2

The reference for Meyer classification was added.

3

spelling/typing mistakes are corrected.

- Abstract, page 2, line 13: "..he..." should be replaced by "...the...".
- Case Presentation, page 4, line 1: "..he..." should be replaced by "...the..."